Author's response to reviews

Title: Comparative psychometric analyses of the SCL-90-R and its short versions in patients with affective disorders

Authors:

Ulrich Prinz (prinz@uni-bremen.de)
Detlev O Nutzinger (nutzinger@psychosomatik.uni-luebeck.de)
Holger Schulz (schulz@uke.de)
Franz Petermann (fpetermann@uni-bremen.de)
Christoph Braukhaus (praxis@dr-braukhaus.de)
Sylke Andreas (sandreas@uke.de)

Version: 2 Date: 3 August 2012

Author's response to reviews: see over
Submission of a manuscript “Comparative psychometric analyses of the SCL-90-R and its short versions in patients with affective disorders”

2nd of August 2012

Dear Prof Olino,

Many thanks for your remarks. Please find our comments on the changes we made.

<table>
<thead>
<tr>
<th>Your comment</th>
<th>Our Answer</th>
</tr>
</thead>
<tbody>
<tr>
<td>Please state the name of ethics committee that actually approved the study</td>
<td>As the data collection was routinely conducted, the study was approved by the institutional review board of the Schoen Hospital Company. We inserted this in the method section of the manuscript.</td>
</tr>
<tr>
<td>A statement to this effect must appear in the Methods section of the</td>
<td></td>
</tr>
<tr>
<td>manuscript, including the name of the body which gave approval, with a</td>
<td></td>
</tr>
<tr>
<td>reference number where appropriate.</td>
<td></td>
</tr>
</tbody>
</table>

Competing interests: Manuscripts should include a ‘Competing interests’ section. This should be placed after the Conclusions/Abbreviations. Please consider the following questions and include a declaration of competing interests in your manuscript:

Financial competing interests
• In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify. => NO
• Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify. => NO
• Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify. => NO
• Do you have any other financial competing interests? If so, please specify. => NO

Non-financial competing interests
• Are there any non-financial competing interests (political, personal, religious, ideological, academic, etc.)? => NO

We carefully proved all the points. There was no competing interest within the study. We stated this in the manuscript.

Competing interest
The authors declare that they have no competing interests.
| intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify. => NO |
| Authors' information: Please place the Authors' Contributions section after Competing interests. Please check the instructions for authors on the journal website for the correct format to use for Authors' Contributions. |
| We included the Authors contribution section at the end of the manuscript as follows: |

**Authors contribution**

UP carried out the study, run the data analyses and drafted a first version of the manuscript. DON, HS, FP and CB participated in the design of the study and helped to draft the manuscript. SA conceived the study and helped to draft the manuscript. All authors read and approved the final manuscript.

Sincerely,

____________________________
Sylke Andreas
Assoc. Prof., Psychologist